Accelerator Status Report

Run Schedule: Aug 21, 2009 to Dec 23, 2009
Aug. 26, 2009

Injector

· Wien angle 44.3 degrees.

· NEW inverted GUN

· Injector energy 64.12 MeV

· Wien angle 17.5. Wien angle will change to 44.6 degrees.
Accelerator

· The accelerator energy was ~0.35% low. Yesterday (8/25) the accelerator energy was raise by ~0.35%. Day and swing shifts were spent to tune the machine. During the tuning it was found that two sextupoles in Arc1 were not at zero current which caused XY coupling. The sextupoles were degaussed and there was no the XY coupling afterwards.
· RF issues. Trip rate is ~10 per hour.

· Some shunts mismatch periodically. They may affect beam. 
· VBV3S01A air hose broke twice.

· The flange at MBZ9E02 developed a leak during beam tuning. 

· Lambertson viewer is still very dark. The marks on the viewer are hard to see.

· PSS A chain dropped the injector.

· BSY dropped to restricted access because someone opened a BSY tunnel door from Hall A side. 

· The accelerator is being set up with total energy 5.764 GeV and unbalanced linacs.
· NL E = 590.6 MeV
· SL E = 549.4 MeV

· RF trip rate is low. 
· RSEP8A was down and fixed today.

Hall A---E05-109
· Hall A ion chambers’ settings.
· Energy 1.204

· Pass 1    
Hall B ---Eg1-DVCS  

· Hall B current was not stable. Reza optimized the injector which stabilized Hall B current. Hall B beam positions at 2C24A were different than expected. The hall is still investigating the problem.

· Current lock does not work well.

· Beam to Hall B Tagger dump tonight (8/19)
· Energy 5.764 Gev

· Pass 5
Hall C---HKS
· Pass 2

· Energy 2.344 GeV

· The HKS is still in the commissioning stage. Up to 15uA CW beam was sent to Hall C. 

· Hall C beam jitters horizontally. The first chicane magnet MDZ3H01is suspected to have an unstable power supply.

· 3H03A and 3H03B harps do not work well.

· New viewer configuration for the hall. Be familiar with the video switcher:
3H02

3H03

Target (upstream the splitter)

3H04B (downstream the splitter)

3H05(YAG)

9D00(dump)
· Commission HKS Thursday swing shift (8/20)
